Author's response to reviews

Title: 23 Gauge Pars Plana Vitrectomy For The Removal Of Retained Intraocular Foreign Bodies

Authors:

Kemal Yuksel (drkemal68@gmail.com)
Ugur Celik (h.ugurcelik@gmail.com)
Cengiz Alagoz (alagozcengiz@gmail.com)
Okkes Baz (dr.bazokkes@hotmail.com)
Burcu Celik (n.burcuguleryuz@gmail.com)
Ahmet Taylan Yazici (ahmettaylan19@gmail.com)

Version: 7 Date: 18 June 2015

Author's response to reviews: see over
Dear Editor

I have done the necessary corrections on the paper; here you can see all the comments that I have made.

1. Please ensure the manuscript follows formatting instructions; the authors' contributions, competing interests and acknowledgements sections should be moved to the end of the manuscript.

   All corrections were done according to the instructions.

2. Authors' contributions: please confirm whether all authors approved the final version of the manuscript.

   All contributions were written.

3. Please provide the name of the ethics committee that gave approval for your study, and clarify whether the informed consent was oral or written. This information should be included in the methods section.

   The necessary corrections were done.

4. Please remove the letter to editor and clear manuscript files from the additional files.

   The necessary corrections were done.

5. Please remove all highlighting from the final version of the manuscript

   The necessary corrections were done.

Kindest Regards.

Ugur Celik